Author's response to reviews

Title: Comparative study of patients with chronic hepatitis C virus infection due to genotypes 1 and 3 referred for treatment in southeast Brazil

Authors:

Aline G Vigani (aline.vigani@sigmanet.com.br)
Maria HP Pavan (mariahpavan@yahoo.com.br)
Raquel Tozzo (tumoca@uol.com.br)
Eduardo SL Gonçales (dugoncales@hotmail.com)
Adriana F Feltrin (afcfeltrin@yahoo.com.br)
Viviane C Fais (fais@unicamp.br)
Maria SK Lazarini (ernestoreginato@uol.com.br)
Neiva SL Gonçales (neiva@unicamp.br)
Fernando L Gonçales Jr (figjr@uol.com.br)

Version: 4 Date: 21 November 2008

Author's response to reviews: see over
To
The BioMed Central Editorial Production Team

MS: 9079354322028216
Comparative Study Among Patients with Chronic Hepatitis C Genotypes 1 and 3 Referred to Treatment.
Aline Vigani, Maria Pavan, Raquel Tozzo, Eduardo Gonçales, Adriana Feltrin, Viviane Fais, Maria SK Lazarini, Neiva Gonçales and Fernando Gonçales Jr

Formatting changes requested

-----------------------------

Major Revisions

Equal contributors “These should be indicated on the title page.

- We didn’t understand what this means.

Abstract - Abstracts must be structured into Background, Methods, Results, Conclusions. Please remember to also update the Abstract details on the submission page.

- The abstract was be structured

Manuscript sections - Manuscript sections should include (in the following order): Abstract; Background; Methods; Results; Discussion; Conclusions; Abbreviations (if any); Competing interests; Authors’ contributions; Acknowledgements; References; Figure legends (if any); Tables (if any); Description of Additional files (if any).

- The manuscript section was included

References - References should be cited in the text using consecutive numbers in square brackets. Please do not use superscript numbers. Also, please place the reference before the full stop. For example ÅeÅ”-text [1].ÅeÅ”ÅT

- References was be cited in the text using consecutive numbers in square brackets.

Minor Revisions

Title - Please use sentence case in the title, both in the manuscript file and the submission system i.e. remove all unnecessary capitals (capitalise only the first word, and proper nouns, and no full stops/periods). For example you would change ÅeÅ”-The Title of My Prestigious Publication: A Scientific Study ÅeÅ”ÅT. To ÅeÅ”-The title of my prestigious publication: a scientific study ÅeÅ”ÅT

- The title was corrected

Email address- please include email addresses for all the authors on the title page. Please use the following format-
This information was included

We look forward to any additional discussion required to address your editorial concerns and look forward to hearing from you soon.

Sincerely,
Aline Gonzalez Vigani, MD

Aline Gonzalez Vigani, MD
Departamento de Clínica Médica
Hospital de Clínicas, UNICAMP
Caixa Postal 6198
Campinas, SP, 13083-970, Brasil
Phone +55 19 3521 7013; FAX +55 19 3289 4107
E-mail: aline.vigani@sigmanet.com.br